\documentstyle[12pt]{report}
\pagestyle{myheadings}
\def\headlineaux{\sc Math 596/688 U: Operating Systems
\hrulefill
}
\markboth{\headlineaux}{\headlineaux}

\newcommand{\assignmentNumber}{3}
\newcommand{\outdate}{10 February, 1994}
\newcommand{\duedate}{17 February, 1994}
\newtheorem{theorem}{Theorem}

\begin{document}
\begin{titlepage}\begin{centerline}{\Huge \framebox{Burt Rosenberg}}
\end{centerline}\end{titlepage}

\section*{Problem Set \assignmentNumber \hfill{\parbox{3in}
    {\small\sc\begin{flushright} Out: \outdate
     \\ Due: \duedate\end{flushright}}}}

\subsection*{Reading}
Please read Chapter 3 in Tanenbaum's {\em Modern
Operating Systems.}

\subsection*{Assignment}
From the problems to Chapter 3, pages 141--144, do problems:
\begin{itemize}
\item 1 through 4,
\item 18 through 22.
\end{itemize}

\subsection*{Midterm Announcement}
The midterm paper will be due Thursday, March 10-th.

\end{document}

